Author’s response to reviews

Title: Household exposure to violence and human rights violations in western Bangladesh (II): history of torture and other traumatic experience of violence and functional assessment of victims

Authors:

Shr-Jie Wang (sjw@rct.dk)
Mohammad Akramul Haque (akramul2000@yahoo.com)
Saber-ud-Daula Masum (liton.masum@yahoo.com)
Shuvodwip Biswas (shuvo85@yahoo.com)
Jens Modvig (jmo@rct.dk)

Version: 8 Date: 19 November 2009

Author’s response to reviews:

Dear editor,

We appreciate for your patience and assistance. We have checked over the manuscript carefully and made typographical changes as follows:

We removed the word “a” from the first line on page 3.

We replaced a set of dash marks with a colon in the third line of the third paragraph on page 6.

We corrected two typographical errors in the second line of the second paragraph on page 13.

We deleted the phrase “in this area” in the second line of the last paragraph on page 18.

We replaced the phrase “Operational Spider Web” with the phrase “Operation Spider Web” in the second line of the second paragraph on page 19.

We corrected a typographical error in the third line of the third paragraph on page 19 by replacing the word “reason” with “season.”

We corrected a typographical error in the seventh line of the second paragraph on page 23 by replacing the word “restoring” with “restore.”

We removed the percentage (0.4) from the column entitled “missing” in the table 4 on page 36.

We corrected two spelling errors in the words “stations” and “military” in Figure 2a.

We replaced “Jama” with “JAMA” in the list of references.
Best regards,

Shr-Jie Sharlenna Wang, MPH, Ph.D.
Public Health Researcher, Epidemiologist
Rehabilitation and Research Centre for Torture Victims
Borgergade 13 P.O. Box 2107
DK-2014 Copenhagen, Denmark
Phone:+45-369-38-638
Fax:+45-337-60510
E-mail: sjw@rct.dk